Author's response to reviews

Title: Use of smokeless tobacco among groups of Pakistani medical students - a cross sectional study

Authors:

Sardar Z Imam (zakariya.imam@gmail.com)
Haq Nawaz (haq.shahzad@gmail.com)
Yasir J Sepah (jamalyasir@gmail.com)
Aqueel H Pabaney (aqueel_pabaney@hotmail.com)
Mahwish Ilyas (mahwish06@gmail.com)
Shehzad Ghaffar (shehzad.ghaffar@aku.edu)

Version: 6 Date: 31 August 2007

Author's response to reviews:

1. Please ensure the authors' names are identical in the manuscript and the submission system. We note a discrepancy, for example Yasir Jamal Sepah/Yasir J Sepah.

This has been done as suggested.

2. Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names.

Can we also ask you to link each author to their corresponding affiliation with the use of superscript numbers.

Affiliation has been written and linked to each author as suggested.

3. Please denote the corresponding author in the authors list, we suggest using *

This has been done as suggested.

4. Please remove the semicolon after conclusions (Conclusions: --> Conclusions)

The colon in front of the heading for the "conclusions section" has been removed as suggested.